Author's response to reviews

Title: Methicillin-resistant Staphylococcus aureus prevalence: Current susceptibility patterns in Trinidad.

Authors:

  Fitzroy A Orrett (drfao4301@yahoo.com)
  Michael Land (land@nsula.edu)

Version: 5 Date: 4 May 2006

Author's response to reviews: see over
May 4, 2006

Dr. Sally Simpson  
The BioMed Central Editorial Team

Dear Sir/Madam

Re: MS: 1737986325930478 - Methicillin-resistant *Staphylococcus aureus* prevalence: Current susceptibility patterns in Trinidad.

Thank you for your e-mail and the “Core-pre-accepting” of the manuscript, in principle.

Corrections/additions made as per request:

1. Titles removed from authors’ names.
2. Capitals (upper cases) removed from title and elsewhere in text, where appropriate.
3. Conclusion added.

Sincerely yours

Dr. Fitzroy A. Orrett.